Child Labor and Resistance to Change Giorgio Bellettini, Carlotta Berti Ceroni and

Gianmarco I.P.Ottaviano NOTA DI LAVORO 79.2003

SEPTEMBER 2003

KNOW – Knowledge, Technology, Human Capital

Giorgio Bellettini, University of Bologna Carlotta Berti Ceroni, University of Bologna Gianmarco I.P. Ottaviano, University of Bologna and CEPR

This paper can be downloaded without charge at:

The Fondazione Eni Enrico Mattei Note di Lavoro Series Index: http://www.feem.it/web/activ/_wp.html

Social Science Research Network Electronic Paper Collection: http://papers.ssrn.com/abstract_id=XXXXXX

The opinions expressed in this paper do not necessarily reflect the position of Fondazione Eni Enrico Mattei

Child Labor and Resistance to Change

Summary

We study the interactions between technological innovation, investment in human capital and child labor. In our setting new technologies require new skills and new skills can be developed only through schooling. In a two-stage game, first firms decide on innovation, then households decide on education. In equilibrium the presence of inefficient child labor depends on parameters related to technology, parents' altruism and the diffusion of firm property. When child labor exists, it is due to either firms reluctance to innovate or households' unwillingness to educate or both. The optimal policy to eliminate child labor depends crucially on its underlying cause. We show that, in some cases, compulsory schooling laws or a ban on child labor are welfare reducing, while a subsidy to innovation is the right tool to eliminate child labor and increase welfare.

Keywords: Child labor, Innovation

JEL: J24, O31

The authors would like to thank Jean Marc Tallon and seminar participants at EUREQUA for useful comments. All remaining errors are the authors'.

Address for correspondence:

Carlotta Berti Ceroni Dipartimento di Scienze Economiche Università di Bologna P.zza Scaravilli 2 Bologna 40126 Italy E-mail: bceroni@economia.unibo.it

1 Introduction

According to International Labor Organization statistics (see ILO [11]), in 1995 at least 120 millions of children between five and twelve years of age worked full time in paid jobs, mostly in underdeveloped regions. This figure would at least double if we took into account children who work part-time. These impressive statistics awake concern both in developing and developed countries and command theoretical as well as policy-oriented analysis. In June of 1999 the 174 member nations of the ILO passed an international convention, upon ratification of which member states are pledged to eliminate the worst forms of child labor immediately and to end, in the long run, all forms of child labor.

A central question for national governments and international organizations is therefore how to intervene in order to reduce the extent of the phenomenon. Indeed, from the point of view of economic theory, public intervention should be justified either by efficiency arguments (for example, in the presence of some kind of externalities or coordination failures) or by social preferences that call for some form of redistribution in favor of children and possibly poor households. Moreover, whatever the justification is, it is clear that any government intervention aimed at reducing or eliminating child labor will affect the current and future welfare of children and is likely to have relevant spillovers on others. It is therefore very important to better understand the determinants of child labor and design policy interventions upon careful analysis and research.

Since the publication of a recent paper by Basu and Van [3], there has been a surge of interest among economists for the theoretical investigation of the causes of child labor aiming at assessing the desirability and efficacy of alternative remedies. However, almost all contributions have so far dealt with the determinants of the supply of (child) labor, focusing mainly on the role of poverty, fertility and liquidity constraints.¹ Furthermore, many of these contributions fail to give a clear-cut argument on the efficiency properties of child labor and the welfare effects of government intervention in this area.

This paper contributes to the literature on the economics of child labor by focusing on the analysis of the determinants of the firm's demand for unskilled (child) labor versus more skilled labor and its interaction with the educational choices of households. In fact, it is widely recognized by economic historians that in many circumstances the reduction or elimination of child labor has been driven by technological progress and the substitution of adult and educated workers for young and unskilled workers (for historical examples,

¹See for example Baland and Robinson [1], Bellettini and Berti Ceroni [4], Cigno and Rosati [5], Dessy [6], Ranjan [14] and Swinnerton and Rogers [15], [16]. An exhaustive survey of the literature on the economics of child labor can be found in Basu [2].

see Galbi [9], Levy [12] and Weiner [17]).

In order to tackle these issues, we develop a model that studies the relationship between technological innovation, child labor and investment in human capital. In a two-period economy, we analyze the joint decisions of altruistic parents (who must decide whether to send their children to work or to school) and firms (which must decide whether to switch to a new technology or not). These decisions are closely interrelated since the returns to education depend on the level of technology and, in turn, the profitability of innovation depends on the quality and skills of the labor force.

More specifically, we assume that technological innovation is decided by the owners of a monopolistic firm. Innovation entails the adoption of a new vintage technology with lower marginal costs of production. However, the adoption of the new vintage implies that the old one becomes freely available to competitors². Moreover, for the new vintage to become operative it is necessary that the labor force acquires the adequate skills or, in other words, that children go to school.

The equilibrium of the model is defined as the outcome of a sequential game where the owners of the monopolistic firm ('capitalists') move first and decide simultaneously whether to innovate or not and whether to send their children to school or work. After observing the capitalists' move, workers decide whether to send their own children to school or work. We show that our economy always displays a unique subgame perfect equilibrium, whose features depend on the value of parameters related to the technology, the degree of parents' altruism and the degree of diffusion of firm's property. When child labor appears in equilibrium, that is due to either firm's reluctance to innovate or workers' unwillingness to educate or both.

Focusing on situations in which child labor is inefficient, we argue that the adequate policy intervention (whether compulsory schooling laws, ban on child labor, subsidies to innovation or to education, etc.) depends crucially on the underlying causes. In particular, we point out circumstances in which the introduction of compulsory schooling laws or a ban on child labor are both welfare reducing, while a subsidy to innovation succeeds in both eliminating child labor and increasing welfare.

Recent work by Dessy and Pallage [7] also stresses the complementarity between simultaneous decisions on technology adoption and skill accumulation. In their case such

 $^{^{2}}$ To justify this assumption we could imagine that the information contained in the patent protecting the state-of-the-art technology allows competitors to imitate the previous vintage technology without breaking the patent. In a different context, Dinopoulos and Segerstrom [8] make the similar assumption that previously patented products can be competitively produced by firms when further innovation occurs in the same industry.

complementarity gives rise to a multiplicity of Pareto-rankable Nash equilibria so that a coordination failure between firms and households may support a suboptimal outcome with technological stagnation and child labor. In our case, multiple equilibria are not part of the story. On purpose, they are avoided by the sequential structure of the game: there is always a unique subgame perfect equilibrium. Accordingly, self-fulfilling 'bad' expectations are not the cause of child labor. Consequently, we are able to push our policy analysis beyond helping 'good' expectations to coordinate.

The rest of the paper is organized as follows. Section 2 introduces the model. Section 3 characterizes the efficient outcome. Section 4 describes the equilibrium outcome. Section 5 presents comparative statics results. Section 6 discusses policy implications and section 7 concludes.

2 The model

We consider a two-period economy, populated by a continuum of identical households of measure 1. In the first period (period t) each household consists of 1 adult and $n \ge 1$ children for a total of n+1 individuals. Adults live for one period, children for two, so that only children are alive in the second period (period t+1) and the household dimension shrinks to n. Each individual is endowed with 1 unit of time per period. While adults supply their units of time inelastically to the labor market, children can choose whether to work or to go to school. The decision on children's time use is made by their altruistic parents. Education is free of charge so that its only (indirect) cost is the foregone labor income of children. The benefit of education comes from human capital accumulation that, in the second period, will allow grown-up children to operate more sophisticated technologies.

On the demand side, due to altruism, the preferences of a typical adult (and thus of the corresponding household) are represented by an intertemporal utility function defined over the consumption of a unique homogeneous good. Specifically, utility is assumed to be linear in the amounts consumed in the two periods:

$$U = (1+n)c_t + \rho n c_{t+1} \tag{1}$$

where c_t is consumption at time t and $\rho \in (0, 1)$ is the discount factor. There are no capital markets so that in each period expenditures equal income.³

 $^{^{3}}$ Given linear utility, in our model the absence of capital markets is not the source of child labor but only a simplifying (realistic) assumption. For an analysis in which child labor arises due to credit constraints see Ranjan [14].

On the supply side, we adopt the dual-technology model of Murphy, Shleifer, and Vishny [13] and we adapt it to our intertemporal setting. In the first period, two technologies, 'traditional' and 'modern', are available for the production of the unique consumption good. Both technologies are linear and use labor as their only input. They differ, however, in terms of productivity, the modern technology being more productive than the traditional one. While the traditional technology is freely disposable, the modern one is exclusive property of one firm only. Due to free disposal, when active, the traditional technology is operated by perfectly competitive firms that are publicly and equally owned by all households. On the contrary, the ownership of the modern firm is equally shared only by a fraction θ of the households ('capitalists'), with $\theta \in [0, 1]$.

Specifically, we assume that, when employed in the traditional technology, one unit of labor produces α_0 units of good, while it produces $\alpha_1 > \alpha_0$ units of good when employed in the modern one. Choosing labor as numeraire, this implies that in the former case the marginal cost of production is $1/\alpha_0$ while in the latter it is $1/\alpha_1 < 1/\alpha_0$.

Profit maximization leads the modern firm to exploit its cost advantage in order to prevent traditional firms from entering the market at all. In particular, this is achieved through limit pricing, that is, by setting a price at which traditional firms are just indifferent between operating or not. That implies that in equilibrium the modern firm's price equals the marginal cost of traditional firms:

$$p_t = 1/\alpha_0 \tag{2}$$

The result is, thus, that the modern firm operates as a monopolist and earns nominal profits:

$$\Pi_t = (p_t - 1/\alpha_1)y_t \tag{3}$$

where y_t is its output and the term between parentheses is the price-cost margin. In real terms, profits can be rewritten as:

$$\pi_t = \frac{\Pi_t}{p_t} = \frac{1/\alpha_0 - 1/\alpha_1}{1/\alpha_0} y_t = \frac{\alpha_1 - \alpha_0}{\alpha_1} y_t \tag{4}$$

This leads to aggregate nominal and real incomes that are respectively:

$$Y_t = p_t y_t = (1/\alpha_0) y_t = \Pi_t + N_t$$
(5)

$$y_t = \pi_t + \alpha_0 N_t \tag{6}$$

where N_t is the total number of individuals (adults plus children) who work at time t.

Solving the system formed by (4) and (6), we obtain the following expression for profits and income in real terms:

$$\pi_t = (\alpha_1 - \alpha_0) N_t \tag{7}$$

$$y_t = \alpha_1 N_t \tag{8}$$

In the first period the monopolist, and thus the capitalists, also face an intertemporal decision in that they can decide whether to update the modern technology or not. If they do, the 'updated' technology will be ready for production in the second period with productivity $\alpha_2 > \alpha_1 > \alpha_0$. However, its operation by grown-up children will be possible only if they have attained an adequate level of human capital through education in the first period. In other words, the capitalists as well as the other households (henceforth called simply 'workers') face a second intertemporal decision on whether to educate their children or not and the updated technology will be implementable only if the innovation decision of the monopolist is matched by the education decision of the households.

More precisely, we assume that the capitalists move first in anticipation of households' choices. The exact timing of events is the following. At the beginning of period t, the capitalists as a group decide whether to update (I) or not (NI) as well as whether to send their children to school (E^o) or not (NE^o) . Then, the workers decide whether to send their children to school (E) or not (NE). After that, during period t, production, education (if any) and consumption take place. Finally, at the end of period t, adults pass away. At the beginning of period t+1, if capitalists have previously decided to update, the new technology becomes available to the monopolist only while the outdated one becomes freely available. Lastly, production and consumption by grown up children take place.

3 The efficient outcome

We are interested in investigating child labor as a market failure. To do so, we need to assess under which conditions child labor is inefficient. Our welfare measure is the sum of the indirect utilities of all capitalits and workers. Given linear utility this corresponds to the present value of aggregate output $y_1 + \rho y_2$.

Straightforward calculations show that:

Proposition 1 Aggregate welfare is maximized at either (I, E^o, E) or (NI, NE^o, NE) . The former dominates the latter if and only if

$$\alpha_2/\alpha_1 > 1 + 1/\rho \tag{9}$$

Intuitively, a planner would choose to send all children to school and to update the technology whenever the present value of the increase in future output due to innovation offsets the current loss of production due to education. Notice that the parameter α_0 is irrelevant for the planner's choice because it affects the distribution of income but it does not affect the output increase.

In what follows we assume (9) to hold, that is, we focus on situations in which, when child labor emerges in equilibrium, it is inefficient so that there is scope for welfareimproving policy intervention.

4 The equilibrium

We are now ready to investigate the conditions under which the market outcome is caracterized by inefficient child labor. Specifically, we look for the Subgame Perfect Equilibrium (SPE) of the two-stage game of complete and perfect information between capitalists and workers. As we discussed above, capitalists move first. Their chosen action will be denoted by a^o . After having observed the action of the capitalists, workers choose their own favorite action a. Clearly, $a^o = (a_1^o, a_2^o)$ where $a_1^o \in A_1^o$, $a_2^o \in A_2^o$, $A_1^o = \{I, NI\}$ and $A_2^o = \{E^o, NE^o\}$. Similarly, $a \in A$ where $A = \{E, NE\}$.

Figure 1 summarizes the tree of the strategic interaction between capitalists and workers with the eight possible outcomes. The associated payoffs are the indirect utilities of the two interest groups, capitalists (labeled again by °) and workers (no label), calculated from (1). To derive them, the crucial thing to keep in mind is that for the updated technology to be operative in the second period, some children must attend school in the first period, in which case their current real labor income $(\alpha_0 n)$ is traded against higher future real labor income $(\alpha_1 n)$. Obviously, this also implies trading off current real profit against higher future real profit for the monopolist, since children attending school are withdrawn from production in the first period. The eight payoffs of workers (A, B, C, D, E, F, G, H, K) and the eight payoffs of capitalists $(A^o, B^o, C^o, D^o, E^o, F^o, G^o, H^o, K^o)$ are reported in the Appendix.



Figure 1: The extensive-form representation of the game

4.1 The choice of workers

Let us characterize the SPE by backward induction taking into account the extensive form depicted in Figure 1. After observing capitalists' decisions regarding innovation and education, workers must decide whether to send their children to school (a = E) or not (a = NE). Here we can distinguish between three cases.

1. Given $a^o = (NI, E^o)$ or $a^o = (NI, NE^o)$ workers' optimal response is a = NE since C > A and D > B. Intuitively, when capitalists do not innovate, the future real wage is equal to the current one (α_0) , since the output price remains constant $(1/\alpha_0)$. Workers have therefore no incentive to send children to school, since this would imply a loss in current labor income against a zero gain in future labor income.

2. Given $a^o = (I, NE^o)$, workers' optimal response is a = NE since F < H. If capitalists innovate but do not send their children to school, they will be unable to operate the up-dated technology unless workers send their children to school. However, workers have no incentives to do so, since they anticipate that the future real wage will be higher in any case. The reason is the following. If workers' children get educated at time t, the monopolistic firm will be able to operate the updated technology at time t + 1 and set the product price at $1/\alpha_1$, so that the real wage will be equal to α_1 . Otherwise, the monopolistic firm will be unable to operate the up-dated technology at time t + 1. Nonetheless, since innovation makes the modern technology freely disposable, a competitive fringe of firms will step in using that technology and the real wage will be still equal to α_1 . 3. Given $a^o = (I, E^o)$, workers' optimal response is a = E if and only if K > G, that is if and only if $\rho \alpha_1 > \alpha_0$. If capitalists innovate and send their children to school they will operate the updated technology at time t + 1 no matter what workers do. In order to decide whether to send their children to school or work at time t + 1, workers must compare the loss in terms of current labor income (opportunity cost of education), α_0 , with the discounted net gain in terms of future labor income, which is given by $\rho \alpha_1$ since in this case only educated workers can be employed at time t + 1.

Thus, for workers to be willing to send their children to school it must be $\rho \alpha_1 > \alpha_0$.

4.2 The choice of capitalists

We can now turn to the characterization of capitalists' best course of action, given workers' best responses. Capitalists must take two simultaneous decisions: whether to adopt the new vintage technology $(a_1^o = I)$ or not $(a_1^o = NI)$ and whether to send their children to school $(a_2^o = E^o)$ or not $(a_2^o = NE^o)$.

It is immediate to verify that, anticipating workers' best responses, capitalists always prefer $a^o = (NI, NE^o)$ to $a^o = (NI, E^o)$, since $D^o > C^o$ and $B^o > A^o$. Intuitively, if capitalists decide not to update the modern technology, in their role of workers they will have no incentives to send their children to school, since this would imply a loss in terms of current labor and profit income against a zero gain in terms of future labor and profit income.

By the same token, $a^o = (NI, NE^o)$ is preferred to $a^o = (I, NE^o)$ since $D^o > H^o$. As we know, if capitalists innovate but do not send children to school, workers will not send children to school either so that capitalists will be unable to operate the new technology in period t+1 but, at the same time, also to prevent the entry of the competitive fringe. For $\theta < 1$, if the capitalists send their children to school, the ensuing loss in terms of discounted profit income, $\rho(\alpha_1 - \alpha_0) n$, is always larger than the gain in terms of discounted wage income, $\theta \rho(\alpha_1 - \alpha_0) n$.

Therefore, we can conclude that

Remark 2 In equilibrium children never attend school in the absence of innovation and innovation is impossible unless some children go to school.

Once we have eliminated actions $a^o = (NI, E^o)$ and $a^o = (I, NE^o)$, we can finally analyze the capitalists' choice between action $a^o = (I, E^o)$ and $a^o = (NI, NE^o)$. Different scenarios emerge depending on workers' willingness to send their children to school, that is depending on the relation between $\rho \alpha_1$ and α_0 . 1. When $\rho \alpha_1 > \alpha_0$, capitalists prefer $a^o = (I, E^o)$ to $a^o = (NI, NE^o)$ if and only if $K^o > D^o$, that is if and only if:

$$(1/\theta)[\rho(\alpha_2 - \alpha_1) - (1+\rho)(\alpha_1 - \alpha_0)]n + [\rho\alpha_1 - (1+\rho)\alpha_0]n > 0$$
(10)

In words, if the discounted future income gain in terms of higher profits, that is $\rho [(\alpha_2 - \alpha_1) - (\alpha_1 - \alpha_0)] n$, and higher wages, $\rho (\alpha_1 - \alpha_0) \theta n$, is larger than the opportunity cost of innovation in terms of current income $[(\alpha_1 - \alpha_0) + \theta \alpha_0] n$, capitalists will innovate and send their children to school. As we know, this in turn implies that workers will also send children to school, so that (I, E^o, E) would be the SPE of the game.

If condition (10) is not satisfied, capitalists do not innovate and nobody sends children to school, so that (NI, NE^o, NE) is the SPE of the game. Notice that in this case child labor emerges as a consequence of capitalists' resistance to innovation as well as to education and hurts workers who would have been better off had the firm innovated.

2. When $\rho \alpha_1 < \alpha_0$ (i.e. when workers never send children to school), capitalists prefer $a^o = (I, E^o)$ to $a^o = (NI, NE^o)$ if and only if $G^o > D^o$, that is if and only if:

$$(1/\theta)[\rho\theta(\alpha_2 - \alpha_1) - (\theta + \rho)(\alpha_1 - \alpha_0)]n + [\rho\alpha_1 - (1 + \rho)\alpha_0]n > 0$$
(11)

The net gain from profits following innovation is different with respect to the preceding case since only capitalists' children go to school at time t (implying a smaller loss in current profit) and work at time t + 1 (implying a smaller gain in future profits). In this case, the SPE equilibrium of the game would be (I, E^o, NE) , implying innovation and 'partial' child labor: only capitalist offsprings go to school.

If condition (11) is not satisfied, capitalists do not innovate and nobody sends children to school, so that (NI, NE^o, NE) is the SPE of the game. In this case, both workers and capitalists resist education.

In order to summarize the above results, it is useful to provide a graphical representation of the possible equilibrium outcomes. Denote with EE, II and IE^{o} the loci defined respectively by the following equations:

$$\frac{\alpha_1}{\alpha_0} = \frac{1}{\rho} \tag{12}$$

$$\frac{\alpha_2}{\alpha_1} = 1 - \theta + \frac{1+\rho}{\rho} - (1-\theta)\frac{1+\rho}{\rho}\frac{1}{\alpha_1/\alpha_0}$$
(13)

$$\frac{\alpha_2}{\alpha_1} = \frac{\rho + \theta}{\rho \theta} - \frac{1 - \theta}{\theta} \frac{1}{\alpha_1 / \alpha_0}$$
(14)

in the $(\alpha_1/\alpha_0, \alpha_2/\alpha_1)$ space. Notice that II and IE^o are derived by equations (10) and (11). Figure 2 depicts the three loci for $0 < \theta < 1$ and $\theta + \rho < 1$ in the relevant area defined by (9).



Figure 2: The equilibrium $(0 < \theta < 1)$

By inspection of Figure 2, we can restate the above results as follows:

Proposition 3 The II, EE and IE° loci identify four regions. To the left of EE, there is partial child labor (only capitalists' children go to school) above the IE° locus (region NW) and complete child labor below it (region SW). To the right of EE, there is no child labor above the II curve (region NE) and complete child labor below it (region SE).

Notice that, while in region SE child labor is the result of firm's resistance to innovation (since the price-cost margin after innovation is low relative to the current one), in region SW both workers and capitalists resist education and innovation (also the wage increase is low). Moreover, even if in region NW the (profit) gain from innovation is high enough to make updating beneficial for the capitalists, wage growth following innovation is not high enough to make education beneficial for the workers. In this case, child labor occurs among workers notwithstanding firm's willingness to innovate.

5 The role of property diffusion

Let us now analyze how the equilibrium characterization depends on the degree of property diffusion of the monopolistic firm θ . Comparative statics results are summarized by Figure 3.

On the one hand, as θ increases, making property more diffuse and each owner's share smaller, the *II* locus rotates clockwise around point $(1+1/\rho, 1+1/\rho)$ and becomes flatter, reaching $\frac{\alpha_2}{\alpha_1} = 1 + 1/\rho$ as θ tends to 1. Ceteris paribus, this makes region *NE* larger (and region *SE* smaller). On the other hand, the *IE*^o locus rotates clockwise around point $(1, 1 + 1/\rho)$ and becomes flatter, also reaching $\frac{\alpha_2}{\alpha_1} = 1 + 1/\rho$ as θ tends to 1. Ceteris paribus, this makes region *NW* larger (and region *SW* smaller).

In order to understand the effect of changes in θ on the *II* locus, notice that (13) can be rewritten as:

$$[\rho\alpha_1 - (1+\rho)\alpha_0] + (1/\theta)[\rho(\alpha_2 - \alpha_1) - (1+\rho)(\alpha_1 - \alpha_0)] = 0$$
(15)

where the first term represents the net discounted gain (or loss) from innovation for capitalist households deriving from wage growth and the second term represents the net gain (loss) deriving from profit growth.

When the first term is negative (positive), an increase in θ implies that a higher (lower) net gain from profit growth is required for the above equation to be satisfied. For example, in region SE innovation favors capitalists in terms of wages, but hurts them in terms of profits. Thus, when the weight of profit income decreases (θ increases), a smaller gain from wage growth is necessary to compensate capitalists for the profit loss. Hence, region SE shrinks as θ increases (given $(\alpha_1 - \alpha_0), (\alpha_2 - \alpha_1)$ must decrease).



Figure 3: The equilibrium (θ grows)

In the limit case where $\theta = 1$ (i.e. the property of the firm is perfectly diffused), the IE^{o} and II loci both collapse to $\alpha_{2}/\alpha_{1} = 1 + 1/\rho$, and the EE locus becomes irrelevant since there are no 'pure' workers. Therefore, innovation always takes place and child labor disappears.

In a different framework, Swinnerton and Rogers [15] [16] also discuss the relationship between firm property diffusion (equality) and child labor incidence. In their analysis, a reduction in inequality can raise poor families income above the subsistence level and permit them to withdraw children from work. Our point is quite different. More diffused property of the firm reduces child labor because it tends to remove the distributional conflict regarding innovation and education between workers and capitalists.

In the limit case where $\theta = 0$, the IE^o locus is no longer relevant and the EE and II loci divide the space in three regions (see Figure 4). An equilibrium with (complete) child labor and no innovation occurs in all regions but one, that is the region to the right of EE and above II (region NE'), where the monopolistic firm innovates and workers send children to school. In region SE', that is to the right of EE and below II, child labor occurs because the monopolist resists innovation (profit after innovation are low relative to current one). In region W, that is to the left of EE child labor occurs because workers are not willing to educate their children (wage growth following innovation is small).



Figure 4: The equilibrium $(\theta = 0)$

6 Policy analysis

Figure 2 reveals that there are three sources of inefficient child labor depending on the combinations of α_1/α_0 and α_2/α_1 . In region NW in equilibrium there is technological update but only capitalists educate their children ('elitarian push'). In region SW, workers' opposition to education forces all children to work ('educational impasse'). In region SE, it is capitalists' opposition to innovation that forces all children to work ('technological impasse'). Therefore, in all three regions there is scope for government intervention.

To analyze government intervention, let us add an initial stage to the previous twostage game. In the initial stage, the government announces its economic policy (to which it is committed) in order to achieve the efficient allocation (I, E^o, E) . In the second and third stage, after being informed about government policy, capitalists and workers choose their optimal actions, in order to maximize their welfare.

As we have discussed above, in each region of Figure 2 the inefficiency of the equilibrium originates from different sources; thus, the appropriate government policy will depend on parameter values. In particular, since in regions NW and SW the inefficiency stems from the fact that workers do not find it worthwhile to educate their children, optimal intervention entails a subsidy to education financed through lump-sum taxation on capitalists (or, equivalently, on profits). Indeed, in both cases, universal education would allow the capitalists to fully reap the benefits of technological innovation, in terms of higher profits.

Differently, in region SE the source of inefficiency is the capitalists' unwillingness to innovate. In this case optimal intervention entails a subsidy to innovation funded through lump-sum taxation on workers (or, equivalently, on wages).

Let us now calculate the minimum subsidy which implements the efficient allocation.

• Region NW - Elitarian push:

Assume that the government announces that it will pay a subsidy to *every* household (whether capitalist or worker) which sends its children to school. The minimum net subsidy received by workers must be such that they are indifferent between outcome (I, E^o, NE) and outcome (I, E^o, E) . Given the payoffs associated to these outcomes, straightforward calculations show that the total subsidy that has to be paid to all households is S = $n(\alpha_0 - \rho\alpha_1)$. Since this subsidy is financed through lump-sum taxation on profits, the *net* amount paid by capitalists (that is, the subsidy received by workers) is $(1 - \theta)n(\alpha_0 - \rho\alpha_1)$. Formally, if we denote the action of the government with a^g , we can write $a^g = (S, T_\pi) =$ $(n(\alpha_0 - \rho\alpha_1), n(\alpha_0 - \rho\alpha_1))$ where T_π is the tax on profit. It is immediate to show that given a^g , the equilibrium outcome of the three-stage game is (I, E^o, E) and every player is at least as well off as in the two-stage game in which the government was inactive.

• Region SW - Educational impasse:

Again, assume that the government pays a subsidy to *every* household which sends its children to school. The net subsidy received by workers must be such that they are indifferent between outcome (NI, NE^o, NE) and outcome (I, E^o, E) . Thus, the total amount of the subsidy to be paid is $S = n(\alpha_0 - \rho\alpha_1 + \rho\alpha_0)$. This subsidy is financed through lump-sum taxation on profits. Thus the *net* amount paid by capitalists is $(1 - \theta)n(\alpha_0 - \rho\alpha_1 + \rho\alpha_0)$. In this case, $a^g = (S, T_\pi) = (n(\alpha_0 - \rho\alpha_1 + \rho\alpha_0), n(\alpha_0 - \rho\alpha_1 + \rho\alpha_0))$. Again, given a^g , the equilibrium outcome of the three-stage game is (I, E^o, E) and every player is at least as well off as in the two-stage game.

• Region SE - Technological impasse:

Now, assume that the government pays a subsidy to capitalists who innovate. The net subsidy received by capitalists must be such that they are indifferent between outcome (NI, NE^o, NE) and outcome (I, E^o, E) . Since this subsidy is financed through lump-sum taxation on wages, it can be easily verified that the total amount of the subsidy to be paid is $S'/(1-\theta)$ where:

$$S' = (\alpha_1 - \alpha_0)(1 + \rho - \theta\rho) + \theta\alpha_0 - \rho(\alpha_2 - \alpha_1).$$
(16)

Once again, given $a^g = (S, T_w) = (S'/(1-\theta), S'/(1-\theta))$, the equilibrium outcome of the three-stage game is (I, E^o, E) and every player is at least as well off as in the two-stage game.

Our previous analysis can be used to evaluate one of the most commonly used form of intervention against child labor, namely compulsory education laws.

On the one hand, in regions NW and SW, the implementation of legislative measures to eliminate child labor provides the firm with the right incentives to innovate.⁴ However, the resulting technological upgrade will benefit the owners of the monopolistic firm at the expenses of the workers who are forced to send their children to school. In this case, without any compensation to workers, legislative measures are not Pareto improving and imply a redistribution of income from workers to capitalists, where the latter enjoy the welfare gain from universal education at zero cost.

From a political economy perspective, the opportunity of such redistribution may justify the political (and economic) support of these measures by the owners of the firms. Weiner [17] discusses a historical example where the owners of the firms were in favor of the introduction of legislative measures against child labor. According to him "in India the proprietors of large businesses have not opposed child-labor laws... One of the complaints of managers of large firms is that their labor force is not sufficiently educated, that

⁴In this paper we abstract from the problem of enforceability and implementability of legislative measures. For an analysis of the role of legislation enforcement in the fight against child labor, see Bellettini and Berti Ceroni [4].

too many workers are unable to read manuals or follow the simple instructions written on machines". Galor and Moav [10] provide several historical examples to argue that in the second phase of the Industrial Revolution "the capitalists were among the prime beneficiaries of the potential accumulation of human capital by the masses", so that they had the incentives to financially support public education.

On the other hand, notice that, in region SE, compulsory schooling legislation alone would bring about a welfare loss for *all* agents, since every child would be forced to go to school but firms would still not find it profitable to innovate. In this case, from a political point of view, legislative measures aimed at reducing the incidence of child labor would be hard to implement in the absence of innovation subsidies.⁵

7 Conclusion

By focusing on poverty and the determinants of the supply of child labor, a recent literature has tried to assess the welfare consequences of government intervention aimed at the reduction or elimination of the phenomenon. This paper adds to this literature by supplementing the analysis with an investigation of the determinants of the demand of the firm for child labor. More specifically, we study the relationship between technological innovation, education, and child labor in a setting where the returns to education depend on the level of technology and the profitability of technological upgrade depends on the quality of the labor force.

Our framework allows us to derive clear-cut welfare implications of public intervention in the area of child labor. In general, we have shown that legislative intervention (that is, child labor bans and compulsory education laws) cannot be Pareto improving unless some kind of redistribution from the owners of the firm to the workers also takes place. Moreover, there are cases where legislation alone decreases the welfare of *all* agents, since it does not provide the right incentives for the firm to innovate. In this case, the optimal policy would be to subsidize technological upgrade.

The simplicity of the framework also shows a potential for fruitful extensions. First, our model could be extended to a multi-period set-up. This would allow one to fully capture the long-run consequences of child labor and public intervention on human capital accumulation and economic development. Second, it could be enriched from a political economy point of view to investigate the role of different institutional mechanisms in mediating the conflicting interests of households and firms.

⁵Indeed, subsidies alone would achieve efficiency even without compulsory schooling legislation.

References

- Baland, J-M. and J. A. Robinson (2000), "Is Child Labor Inefficient?", Journal of Political Economy, 108, 663-679.
- [2] Basu, K. (1999), "Child Labor: Cause, Consequence, and Cure, with Remarks on International Labor Standards", *Journal of Economic Literature*, 37, 1083-1119.
- [3] Basu, K. and P. H. Van (1998), "The Economics of Child Labor", American Economic Review, 88, 412-427.
- [4] Bellettini, G. and C. Berti Ceroni (2000), "Compulsory Schooling Laws and the Cure Against Child Labor", University of Bologna, mimeo.
- [5] Cigno, A. and F. Rosati (2000), "Why Do Indian Children Work, and Is It Bad For Them?", ChilD Working Paper, no.1.
- [6] Dessy, S. E. (2000), "A Defense of Compulsory Measures Against Child Labor", Journal of Development Economics, 62, 261-275.
- [7] Dessy, S.E. and S. Pallage (2001), "Child Labor and Coordination Failures", Journal of Development Economics, 65, 469-476.
- [8] Dinopoulos E. and P. Segerstrom (1999), "A Schumpeterian Model of Protection and Relative Wages", American Economic Review, 89, 450-472.
- [9] Galbi, D. A. (1997), "Child Labor and the Division of Labor in the Early English Cotton Mills", Journal of Population Economics, 10, 357-375.
- [10] Galor, O. and O. Moav (2000), "Das Human Capital", Brown University Working Paper no. 2000-17.
- [11] ILO (1998), Report VI (1): Child Labour: Targeting the intolerable, Geneva.
- [12] Levy, V. (1985), "Cropping Pattern, Mechanization, Child Labor, and Fertility Behavior in a Farming Economy: Rural Egypt", *Economic Development and Cultural Change*, 33, 777-791.
- [13] Murphy, K. M., Shleifer, A. and R. W. Vishny (1989), "Industrialization and the Big Push", *Journal of Political Economy*, 97, 1003-1026.

- [14] Ranjan, P. (2001), "Credit Constraints and the Phenomenon of Child Labor", Journal of Development Economics, 64, 81-102.
- [15] Swinnerton, K. A. and C. A. Rogers (1999a), "The Economics of Child Labor: Comment", American Economic Review, 89(5), 1382-1385.
- [16] Swinnerton, K. A. and C. A. Rogers (1999b), "Inequality, Productivity, and Child Labor: Theory and Evidence", mimeo.
- [17] Weiner, M. (1991), The Child and the State in India: Child Labor and Education Policy in Comparative Perspective, Princeton: Princeton University Press.

Appendix - The payoffs of the game

• Workers' payoffs

 $A \equiv (1 - \theta)\alpha_0(1 + \rho n) \text{ associated to outcome } (NI, E^o, E)$ $B \equiv (1 - \theta)\alpha_0(1 + \rho n) \text{ associated to outcome } (NI, NE^o, E)$ $C \equiv (1 - \theta)\alpha_0[1 + (1 + \rho)n] \text{ associated to outcome } (NI, E^o, NE)$ $D \equiv (1 - \theta)\alpha_0[1 + (1 + \rho)n] \text{ associated to outcome } (NI, NE^o, NE)$ $K \equiv (1 - \theta)(\alpha_0 + \rho\alpha_1 n) \text{ associated to outcome } (I, E^o, E)$ $F \equiv (1 - \theta)(\alpha_0 + \rho\alpha_1 n) \text{ associated to outcome } (I, NE^o, E)$ $G \equiv (1 - \theta)\alpha_0(1 + n) \text{ associated to outcome } (I, E^o, NE)$ $H \equiv (1 - \theta)[\alpha_0(1 + n) + \rho\alpha_1 n] \text{ associated to outcome } (I, NE^o, NE)$

• Capitalists' payoffs:

$$\begin{split} A^{o} &\equiv \theta \alpha_{0}(1+\rho n) + (1+\rho n) (\alpha_{1}-\alpha_{0}) \\ B^{o} &\equiv \theta \alpha_{0}[1+(1+\rho)n] + (\alpha_{1}-\alpha_{0}) [1+(\rho+\theta)n] \\ C^{o} &\equiv \theta \alpha_{0}(1+\rho n) + (\alpha_{1}-\alpha_{0}) [1+(1+\rho-\theta)n] \\ D^{o} &\equiv \theta \alpha_{0}[1+(1+\rho)n] + (\alpha_{1}-\alpha_{0}) [1+(1+\rho)n] \\ K^{o} &\equiv \theta (\alpha_{0}+\rho \alpha_{1}n) + (\alpha_{1}-\alpha_{0}) + \rho (\alpha_{2}-\alpha_{1}) n \\ F^{o} &\equiv \theta \alpha_{0}(1+n) + (\alpha_{1}-\alpha_{0}) (1+\theta n) + \rho (\alpha_{2}-\alpha_{1}) (1-\theta)n \\ G^{o} &\equiv \theta (\alpha_{0}+\rho \alpha_{1}n) + (\alpha_{1}-\alpha_{0}) [1+n(1-\theta)] + \rho (\alpha_{2}-\alpha_{1}) \theta n \\ H^{o} &\equiv \theta \alpha_{0}(1+n) + (\alpha_{1}-\alpha_{0}) (1+n) + \rho \theta \alpha_{1}n \end{split}$$

NOTE DI LAVORO DELLA FONDAZIONE ENI ENRICO MATTEI Fondazione Eni Enrico Mattei Working Paper Series

Our working papers are available on the Internet at the following addresses: http://www.feem.it/Feem/Pub/Publications/WPapers/default.html

http://papers.ssrn.com

SUST	1.2002	K. TANO, M.D. FAMINOW, M. KAMUANGA and B. SWALLOW: Using Conjoint Analysis to Estimate Farmers' Preferences for Cottle Traite in West Africa
ETA	2 2002	Efrem CASTELNUOVO and Paolo SURICO: What Does Monetary Policy Reveal about Central Bank's
LIII	2.2002	Preferences?
WAT	3.2002	Duncan KNOWLER and Edward BARBIER: The Economics of a "Mixed Blessing" Effect: A Case Study of the
		Black Sea
CLIM	4.2002	Andreas LÖSCHEL: Technological Change in Economic Models of Environmental Policy: A Survey
VOL	5.2002	Carlo CARRARO and Carmen MARCHIORI: Stable Coalitions
CLIM	6.2002	Marzio GALEOTTI, Alessandro LANZA and Matteo MANERA: Rockets and Feathers Revisited: An International
		Comparison on European Gasoline Markets
ETA	7.2002	Effrosyni DIAMANTOUDI and Effichios S. SARTZETAKIS: Stable International Environmental Agreements: An
		Analytical Approach
KNOW	8.2002	Alain DESDOIGTS: Neoclassical Convergence Versus Technological Catch-up: A Contribution for Reaching a
		Consensus
NRM	9.2002	Giuseppe DI VITA: Renewable Resources and Waste Recycling
KNOW	10.2002	Giorgio BRUNELLO: Is Training More Frequent when Wage Compression is Higher? Evidence from 11
		European Countries
ETA	11.2002	Mordecai KURZ, Hehui JIN and Maurizio MOTOLESE: Endogenous Fluctuations and the Role of Monetary
		Policy
KNOW	12.2002	Reyer GERLAGH and Marjan W. HOFKES: Escaping Lock-in: The Scope for a Transition towards Sustainable
		<u>Growth?</u>
NRM	13.2002	Michele MORETTO and Paolo ROSATO: The Use of Common Property Resources: A Dynamic Model
CLIM	14.2002	Philippe QUIRION: Macroeconomic Effects of an Energy Saving Policy in the Public Sector
CLIM	15.2002	Roberto ROSON: Dynamic and Distributional Effects of Environmental Revenue Recycling Schemes:
		Simulations with a General Equilibrium Model of the Italian Economy
CLIM	16.2002	Francesco RICCI (I): Environmental Policy Growth when Inputs are Differentiated in Pollution Intensity
ETA	17.2002	Alberto PETRUCCT: Devaluation (Levels versus Rates) and Balance of Payments in a Cash-in-Advance
a	10.000	Economy
Coalition	18.2002	László Á. KÓCZY (liv): The Core in the Presence of Externalities
Theory		
Network	10 2002	
Coalition	19.2002	Steven J. BRAMS, Michael A. JONES and D. Marc KILGOUR (liv): Single-Peakedness and Disconnected
Theory		Coalitions
Network	20.2002	Countions
Coalition	20.2002	Guillaume HAERINGER (liv): On the Stability of Cooperation Structures
I neory		
NELWOIK	21 2002	Emiste CAUALI ADO en d Luis i CIDAOLO. Economic en d Environmental Susteinsbilitas A Demenie Ammerica
NKM	21.2002	<i>Fausto CAVALLARO and Luigi CIRAOLO: <u>Economic and Environmental Sustainability: A Dynamic Approacn</u></i>
CLIM	22 2002	In Insular Systems
CLIM	22.2002	Barbard BUCHNER, Carlo CARRARO, 1807 CERSOSINO and Carmen MARCHIORI. Back to Kyoto? US
CLIM	22 2002	And rear LÖSCHEL and Thoug VIANG THANG: The Economic and Environmental Implications of the US
CLIN	23.2002	Anareus EOSCHER and Enorganativo Enaros. The Eonomic and Environmental Impleations of the OS
FTA	24 2002	Maggio GALEOTTI Louis I MACCINI and Eabio SCHIANTARELLI: Inventories Employment and Hours
	24.2002	Marzio GALEOTTI, Louis 5. MACCINI and real of the Environmental Kurnets Curve Misleading? New Evidence from
CLINI	23.2002	Time Series Data for Germany
FTΛ	26 2002	Adam B. IAFEF. Richard G. NEWELL and Robert N. STAVINS: Environmental Policy and Technological
LIA	20.2002	Change
SUST	27 2002	Unserpt C. COOPER and Giovanni SIGNORELLO: Farmer Premiums for the Voluntary Adoption of
5051	27.2002	Conservation Plans
SUST	28 2002	The ANSEA Network: Towards An Analytical Strategic Environmental Assessment
KNOW	29 2002	Paolo SURICO: Geographic Concentration and Increasing Returns: a Survey of Evidence
ETA	30.2002	<i>Robert N. STAVINS:</i> Lessons from the American Experiment with Market-Based Environmental Policies

NRM	31.2002	Carlo GIUPPONI and Paolo ROSATO: Multi-Criteria Analysis and Decision-Support for Water Management at
NDM	22 2002	the Catchment Scale: An Application to Diffuse Pollution Control in the Venice Lagoon
KNOW	33 2002	A SOUBEYRAN and H STAHN : Do Investments in Specialized Knowledge Lead to Composite Good
ICT (O II	55.2002	Industries?
KNOW	34.2002	G. BRUNELLO, M.L. PARISI and Daniela SONEDDA: Labor Taxes, Wage Setting and the Relative Wage
		Effect
CLIM	35.2002	C. BOEMARE and P. QUIRION (lv): Implementing Greenhouse Gas Trading in Europe: Lessons from
CL D.(26 2002	Economic Theory and International Experiences
CLIM	36.2002	<i>I. HE IENBERG</i> (IV): The Tradable Permits Approach to Protecting the Commons: What Have We Learned?
CLIM	38 2002	C. EISCHER (ly): Multinational Tayation and International Emissions Trading
SUST	39 2002	G SIGNORELLO and G PAPPALARDO: Farm Animal Biodiversity Conservation Activities in Europe under
5051	27.2002	the Framework of Agenda 2000
NRM	40.2002	S.M. CAVANAGH, W. M. HANEMANN and R. N. STAVINS: Muffled Price Signals: Household Water Demand
		under Increasing-Block Prices
NRM	41.2002	A. J. PLANTINGA, R. N. LUBOWSKI and R. N. STAVINS: The Effects of Potential Land Development on
CL DA	42 2002	<u>Agricultural Land Prices</u>
CLIM	42.2002	C. OHL (101): Inducing Environmental Co-operation by the Design of Emission Permits
CLIM	45.2002	J. ETCKMANS, D. VAN REGEMORTER and V. VAN STEENDERGHE (IVI). IS KYOLO FALALLY FLAWED! ALL Analysis with MacGEM
CLIM	44 2002	<u>A ANTOCL and S BORGHESI</u> (lvi): Working Too Much in a Polluted World: A North-South Evolutionary
CLIM	11.2002	Model
ETA	45.2002	P. G. FREDRIKSSON, Johan A. LIST and Daniel MILLIMET (lvi): Chasing the Smokestack: Strategic
		Policymaking with Multiple Instruments
ETA	46.2002	Z. YU (lvi): A Theory of Strategic Vertical DFI and the Missing Pollution-Haven Effect
SUST	47.2002	Y. H. FARZIN: Can an Exhaustible Resource Economy Be Sustainable?
SUST	48.2002	Y. H. FARZIN: Sustainability and Hamiltonian Value
KNOW	49.2002	C. PIGA and M. VIVARELLI: Cooperation in R&D and Sample Selection
Coalition	50.2002	M. SERTEL and A. SLINKO (liv): <u>Ranking Committees</u> , Words or Multisets
Theory		
Network	51 2002	Curris CURPARINU (110). Stable Operations with Esternalities
Theory	51.2002	Sergio CORRARINI (IIV): <u>Stable Organizations with Externatities</u>
Network		
FTA	52 2002	Robert N STAVINS: Experience with Market-Based Policy Instruments
ETA	53.2002	C.C. JAEGER. M. LEIMBACH. C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and
2111	00.2002	<i>R. KLEIN</i> (liii): Integrated Assessment Modeling: Modules for Cooperation
CLIM	54.2002	Scott BARRETT (liii): Towards a Better Climate Treaty
ETA	55.2002	
		Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-
CUCT		Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies
3031	56.2002	Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs
SUST	56.2002 57.2002	Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of
SUST	56.2002 57.2002	Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests
SUST SUST	56.2002 57.2002 58.2002	Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Ensure MISSEEL DT and Arture VILL AUCENCO (hii): Low Con Economics in Transition Environ
SUST SUST SUST SUST	56.2002 57.2002 58.2002 59.2002	Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation?
SUST SUST SUST VOL	56.2002 57.2002 58.2002 59.2002 60.2002	Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together
SUST SUST SUST VOL	56.2002 57.2002 58.2002 59.2002 60.2002	Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary
SUST SUST SUST VOL	56.2002 57.2002 58.2002 59.2002 60.2002	Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union
SUST SUST SUST VOL ETA	56.2002 57.2002 58.2002 59.2002 60.2002 61.2002	Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic
SUST SUST SUST VOL ETA	56.2002 57.2002 58.2002 59.2002 60.2002 61.2002	Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity
SUST SUST SUST VOL ETA PRIV	56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002	Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability
SUST SUST SUST VOL ETA PRIV PRIV	56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002	Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on
SUST SUST SUST VOL ETA PRIV PRIV	56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002	Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q
SUST SUST SUST VOL ETA PRIV PRIV PRIV	56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 64.2002	Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent
SUST SUST SUST VOL ETA PRIV PRIV PRIV	56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 64.2002 64.2002	Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity
SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST	56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 63.2002 63.2002 64.2002 65.2002	Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life
SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST ETA	56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 63.2002 64.2002 65.2002 66 2002	Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economics in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo SURICO: US Monetary Policy Rules: the Case for Asymmetric Preferences
SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST ETA PRIV	56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 63.2002 64.2002 65.2002 65.2002 67.2002	Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economics in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo SURICO: US Monetary Policy Rules: the Case for Asymmetric Preferences Rinaldo BRAU and Massimo FLORIO: Privatisations as Price Reforms: Evaluating Consumers' Welfare
SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST ETA PRIV	 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 61.2002 63.2002 64.2002 65.2002 66.2002 67.2002 	Richard G, NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economics in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo SURICO: US Monetary Policy Rules: the Case for Asymmetric Preferences Rinaldo BRAU and Massimo FLORIO: Privatisations as Price Reforms: Evaluating Consumers' Welfare Changes in the U.K.
SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST ETA PRIV CLIM	 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 61.2002 63.2002 64.2002 65.2002 66.2002 67.2002 68.2002 	Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo SURICO: US Monetary Policy Rules: the Case for Asymmetric
SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST ETA PRIV CLIM CLIM	56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 64.2002 65.2002 66.2002 67.2002 68.2002 69.2002	Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo SURICO: US Monetary Pol
SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST ETA PRIV CLIM CLIM SUST	56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 64.2002 65.2002 66.2002 67.2002 68.2002 69.2002 70.2002	 Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo SURICO: US Monetary Policy Rules: the Case for Asymmetric Preferences Rinaldo BRAU and Massimo FLORIO: Privatisations as Price Reforms: Evaluating Consumers' Welfare Changes in the U.K. Barbara K. BUCHNER and Roberto ROSON: Conflicting Perspectives in Trade and Environmental Negotiations Philippe QUIRION: Complying with the Kyoto Protocol under Uncertainty: Taxes or Tradable Permits? A
SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST ETA PRIV CLIM CLIM SUST	 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002 64.2002 65.2002 66.2002 67.2002 68.2002 69.2002 70.2002 	 Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economies in Transition Pursue Emissions Trading or Joint Implementation? Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic Terms: Dynamic Efficiency Plus Intergenerational Equity Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent Productivity Orley ASHENFELTER and Michael GREENSTONE: Using Mandated Speed Limits to Measure the Value of a Statistical Life Paolo SURICO: US Monetary Policy Rules: the Case for Asymmetric Preferences Rinaldo BRAU and Massimo FLORIO: Privatisations as Price Reforms: Evaluating Consumers' Welfare Changes in the U.K. Barbara K. BUCHNER and Roberto ROSON: Conflicting Perspectives in Trade and Environmental Negotiations Philippe QUIRION: Complying with the Kyoto Protocol under Uncertainty: Taxes or Tradable Permits?<!--</td-->

NRM	72.2002	Philippe BONTEMS and Pascal FAVARD: Input Use and Capacity Constraint under Uncertainty: The Case of
	72 2002	Irrigation
PRIV	/3.2002	Mohammed OMRAN: <u>The Performance of State-Owned Enterprises and Newly Privatized Firms: Empirical</u>
PRIV	74 2002	Evidence from Egypt Mike RURKART Equato PANUNZL and Andrei SHI EIEER: Family Firms
PRIV	75.2002	<i>Emmanuelle AURIOL, Pierre M. PICARD:</i> Privatizations in Developing Countries and the Government Budget
	/0.2002	Constraint
PRIV	76.2002	Nichole M. CASTATER: Privatization as a Means to Societal Transformation: An Empirical Study of
		Privatization in Central and Eastern Europe and the Former Soviet Union
PRIV	77.2002	Christoph LÜLSFESMANN: Benevolent Government, Managerial Incentives, and the Virtues of Privatization
PRIV	78.2002	Kate BISHOP, Igor FILATOTCHEV and Tomasz MICKIEWICZ: Endogenous Ownership Structure: Factors
		Affecting the Post-Privatisation Equity in Largest Hungarian Firms
PRIV	79.2002	Theodora WELCH and Rick MOLZ: <u>How Does Trade Sale Privatization Work?</u>
DDIV	80 2002	Evidence from the Fixed-Line Telecommunications Sector in Developing Economies
PKIV	80.2002	<i>Alberto R. PETROCCT</i> . Government Debi, Agent Helerogeneity and wealth Displacement in a Small Open
CLIM	81 2002	Timothy SWANSON and Robin MASON (lvi): The Impact of International Environmental Agreements: The Case
CLIM	01.2002	of the Montreal Protocol
PRIV	82.2002	George R.G. CLARKE and Lixin Colin XU: Privatization. Competition and Corruption: How Characteristics of
		Bribe Takers and Payers Affect Bribe Payments to Utilities
PRIV	83.2002	Massimo FLORIO and Katiuscia MANZONI: The Abnormal Returns of UK Privatisations: From Underpricing
		to Outperformance
NRM	84.2002	Nelson LOURENÇO, Carlos RUSSO MACHADO, Maria do ROSÁRIO JORGE and Luís RODRIGUES: An
		Integrated Approach to Understand Territory Dynamics. The Coastal Alentejo (Portugal)
CLIM	85.2002	Peter ZAPFEL and Matti VAINIO (lv): Pathways to European Greenhouse Gas Emissions Trading History and
CL D /	06 0000	Misconceptions
CLIM	86.2002	<i>Pierre COURTOIS</i> : Influence Processes in Climate Change Negotiations: Modelling the Rounds
EIA	87.2002	VIIO FRAGNELLI and Maria Erminia MARINA (IVIII): <u>Environmental Pollution Risk and Insurance</u>
	88.2002	Time COESCHI and Timethy M SWANSON (lyiji): Lost Herizons. The noncooperative management of an
LIA	07.2002	evolutionary biological system
ETA	90.2002	Hans KEIDING (lyiji): Environmental Effects of Consumption: An Approach Using DEA and Cost Sharing
ETA	91.2002	Wietze LISE (Iviii): A Game Model of People's Participation in Forest Management in Northern India
CLIM	92.2002	Jens HORBACH: Structural Change and Environmental Kuznets Curves
ETA	93.2002	Martin P. GROSSKOPF: Towards a More Appropriate Method for Determining the Optimal Scale of Production
		Units
VOL	94.2002	Scott BARRETT and Robert STAVINS: Increasing Participation and Compliance in International Climate Change
CL D /	05 0000	Agreements
CLIM	95.2002	Banu BAYRAMOGLU LISE and Wietze LISE: Climate Change, Environmental NGOs and Public Awareness in
CUM	06 2002	International Structure and Reality Matthian CLACHANT: The Political Economy of Emission Tax Design in Environmental Policy
KNOW	90.2002	Kann ARIGA and Giorgio BRUNELLO: Are the More Educated Receiving More Training? Evidence from
KINOW	77.2002	Thailand
ETA	98.2002	Gianfranco FORTE and Matteo MANERA: Forecasting Volatility in European Stock Markets with Non-linear
2111	2002	GARCH Models
ETA	99.2002	Geoffrey HEAL: Bundling Biodiversity
ETA	100.2002	Geoffrey HEAL, Brian WALKER, Simon LEVIN, Kenneth ARROW, Partha DASGUPTA, Gretchen DAILY, Paul
		EHRLICH, Karl-Goran MALER, Nils KAUTSKY, Jane LUBCHENCO, Steve SCHNEIDER and David
		STARRETT: Genetic Diversity and Interdependent Crop Choices in Agriculture
ETA	101.2002	Geoffrey HEAL: Biodiversity and Globalization
VOL	102.2002	Andreas LANGE: Heterogeneous International Agreements – If per capita emission levels matter
EIA	103.2002	Pierre-Andre JOUVET and Walid OUESLATI: <u>Tax Reform and Public Spending Trade-offs in an Endogenous</u>
ETA	104 2002	Arma ROTTASSO and Alessandre SEMPENELLI: Door Ownership Affect Firms' Efficiency? Banal Date
LIA	104.2002	<i>Annu DOTTASSO unu Alessunuro SEMDENELLI. <u>Does Ownersnip Affect Fillis Efficiency: Fallet Data</u> Evidence on Italy</i>
PRIV	105 2002	Bernardo RORTOLOTTI Frank DE JONG Giovanna NICODANO and Ibolva SCHINDELE: Privatization and
1101	100.2002	Stock Market Liquidity
ETA	106.2002	Haruo IMAI and Mayumi HORIE (Iviii): Pre-Negotiation for an International Emission Reduction Game
PRIV	107.2002	Sudeshna GHOSH BANERJEE and Michael C. MUNGER: Move to Markets? An Empirical Analysis of
		Privatisation in Developing Countries
PRIV	108.2002	Guillaume GIRMENS and Michel GUILLARD: Privatization and Investment: Crowding-Out Effect vs Financial
		Diversification
PRIV	109.2002	Alberto CHONG and Florencio LOPEZ-DE-SILANES: Privatization and Labor Force Restructuring Around the
	110 2002	World New divis CUDT (), Dertial Drivertiantian and Firm Derformer (1)
	110.2002	IVARIATION OUT LAT <u>PARUAL PRIVAUZATION and FIRM PERFORMANCE</u> Examples DEGEORGE Dirk IENTER Alberto MOEL and Pater TUEANO: Solling Company Shares to
1 KIV	111.2002	Reluctant Employees: France Telecom's Experience
		Refue with Employees, Figuree Felecom 5 Experience

PRIV	112.2002	Isaac OTCHERE: Intra-Industry Effects of Privatization Announcements: Evidence from Developed and
		Developing Countries
PRIV	113.2002	Yannis KATSOULAKOS and Elissavet LIKOYANNI: Fiscal and Other Macroeconomic Effects of Privatization
PRIV	114.2002	Guillaume GIRMENS: Privatization, International Asset Trade and Financial Markets
PRIV	115.2002	D. Teja FLOTHO: A Note on Consumption Correlations and European Financial Integration
PRIV	116.2002	Ibolya SCHINDELE and Enrico C. PEROTTI: Pricing Initial Public Offerings in Premature Capital Markets:
		The Case of Hungary
PRIV	1.2003	Gabriella CHIESA and Giovanna NICODANO: Privatization and Financial Market Development: Theoretical
		Issues
PRIV	2.2003	Ibolya SCHINDELE: Theory of Privatization in Eastern Europe: Literature Review
PRIV	3.2003	Wietze LISE, Claudia KEMFERT and Richard S.J. TOL: Strategic Action in the Liberalised German Electricity
		Market
CLIM	4.2003	Laura MARSILIANI and Thomas I. RENSTROM: Environmental Policy and Capital Movements: The Role of
		Government Commitment
KNOW	5.2003	Reyer GERLAGH: Induced Technological Change under Technological Competition
ETA	6.2003	Efrem CASTELNUOVO: Squeezing the Interest Rate Smoothing Weight with a Hybrid Expectations Model
SIEV	7.2003	Anna ALBERINI, Alberto LONGO, Stefania TONIN, Francesco TROMBETTA and Margherita TURVANI: The
		Role of Liability, Regulation and Economic Incentives in Brownfield Remediation and Redevelopment:
		Evidence from Surveys of Developers
NRM	8.2003	Elissatos PAPYRAKIS and Reyer GERLAGH: <u>Natural Resources: A Blessing or a Curse?</u>
CLIM	9.2003	A. CAPARROS, JC. PEREAU and T. TAZDATT: North-South Climate Change Negotiations: a Sequential Game
WNOW	10 2002	with Asymmetric Information
KNOW	10.2003	Giorgio BRUNELLO and Daniele CHECCHE, School Quality and Family Background in Italy
CLIM	11.2003	Element CASTELINGOVO ana Marzio GALEOTTI: Learning By Doing vs Learning By Researching in a Model of
KNOW	12 2002	Climate Change Policy Analysis Crimite MAIGNAN, Circumstance OTTAVIANO and Dine DINELLI (.d.), Economic Creatily, Innovation, Cultural
KNUW	12.2003	Carole MAIGNAN, Glanmarco OTTAVIANO and Dino PINELLI (eas.): Economic Growin, Innovation, Cultural
KNOW	13 2003	Carola MAIGNAN Gianmarco OTTAVIANO Dino PINELLI and Erangasco PULLANI (liv): Bio Ecological
KINOW	13.2003	Diversity vs. Socio-Economic Diversity. A Comparison of Existing Measures
KNOW	14 2003	Maddy IANSSENS and Chris STEVAERT (liv): Theories of Diversity within Organisation Studies: Debates and
KINO W	14.2005	Future Trajectories
KNOW	15 2003	Turin RAYCAN LEVENT Fund MASUREL and Poter NUKAMP (lix): Diversity in Entrepreneurshin: Ethnic and
KINO W	15.2005	Female Roles in Urban Economic Life
KNOW	16 2003	Alexandra BITUSIKOVA (lix): Post-Communist City on its Way from Grey to Colourful: The Case Study from
	10.2005	Slovakia
KNOW	17.2003	Billy E. VAUGHN and Katarina MLEKOV (lix): A Stage Model of Developing an Inclusive Community
KNOW	18.2003	Selma van LONDEN and Arie de RUIJTER (lix): Managing Diversity in a Glocalizing World
Coalition	19.2003	Sergio CURRARINI: On the Stability of Hierarchies in Games with Externalities
Theory		
Network		
PRIV	20.2003	Giacomo CALZOLARI and Alessandro PAVAN (lx): Monopoly with Resale
PRIV	21.2003	Claudio MEZZETTI (lx): Auction Design with Interdependent Valuations: The Generalized Revelation
		Principle, Efficiency, Full Surplus Extraction and Information Acquisition
PRIV	22.2003	Marco LiCalzi and Alessandro PAVAN (lx): Tilting the Supply Schedule to Enhance Competition in Uniform-
		Price Auctions
PRIV	23.2003	David ETTINGER (lx): Bidding among Friends and Enemies
PRIV	24.2003	Hannu VARTIAINEN (lx): Auction Design without Commitment
PRIV	25.2003	Matti KELOHARJU, Kjell G. NYBORG and Kristian RYDQVIST (lx): Strategic Behavior and Underpricing in
		Uniform Price Auctions: Evidence from Finnish Treasury Auctions
PRIV	26.2003	Christine A. PARLOUR and Uday RAJAN (IX): <u>Rationing in IPOs</u>
PRIV	27.2003	Kjell G. NYBORG and Ilya A. STREBULAEV (Ix): <u>Multiple Unit Auctions and Short Squeezes</u>
PRIV	28.2003	Anders LUNANDER and Jan-Eric NILSSON (Ix): Taking the Lab to the Field: Experimental Tests of Alternative
DDUV		Mechanisms to Procure Multiple Contracts
PRIV	29.2003	TangaMcDANIEL and Karsten NEUHOFF (Ix): Use of Long-term Auctions for Network Investment
	30.2003	Emiel MAASLAND and Sander ONDERSTAL (IX): <u>Auctions with Financial Externalities</u>
EIA	31.2003	Michael FINUS and Blanca KUNDSHAGEN: <u>A Non-cooperative Foundation of Core-Stability in Positive</u>
KNOW	22 2002	Externally NIU-Coalition Games
	32.2003	Philippa OUIPION: Polotive Ouotoo: Correct Answer to Uncertainty or Cose of Populatory Conture?
F KI V K NOW	33.2003	Giuganna MEDA Claudio PICA and Donald SIECEL: On the Delationship between D&D and Draductivity.
MINOW	34.2003	Treatment Effect Analysis
ET 4	35 2003	Alessandra DEL ROCA Marrie GALEOTTL and Paola ROTA: Non-convertises in the Adjustment of Different
LIA	55.2005	Canital Inputs: A Firm-level Investigation
		Cupitul Inputs, 111 Inn 19491 Involuguion

GG PRIV	36.2003 37.2003	Matthieu GLACHANT: Voluntary Agreements under Endogenous Legislative Threats Narjess BOUBAKRI, Jean-Claude COSSET and Omrane GUEDHAMI: <u>Postprivatization</u> Corporate Coversance: the Pole of Ownership Structure and Investor Protection
CUM	28 2002	Devertance, the Kore of Ownersing Studente and Investor Protection
CLIM	38.2003	Roly GOLOMBER and Michael HOLE. Climate Policy under Technology Spilovers
KNOW	39.2003	Sim BEN TOUSSEF: Transboundary Poliution, R&D Spillovers and International Trade
CIN	40.2003	Carlo CARRARO and Carmen MARCHIORI: Endogenous Strategic Issue Linkage in International Negotiations
KNOW	41.2003	Sonia OREFFICE: Abortion and Female Power in the Household: Evidence from Labor Supply
KNOW	42.2003	Timo GOESCHL and Timothy SWANSON: On Biology and Technology: The Economics of Managing
		Biotechnologies
ETA	43.2003	<i>Giorgio BUSETTI and Matteo MANERA</i> : <u>STAR-GARCH Models for Stock Market Interactions in the Pacific</u> <u>Basin Region, Japan and US</u>
CLIM	44.2003	Katrin MILLOCK and Céline NAUGES: The French Tax on Air Pollution: Some Preliminary Results on its Effectiveness
PRIV	45.2003	Bernardo BORTOLOTTI and Paolo PINOTTI: The Political Economy of Privatization
SIEV	46.2003	Elbert DIJKGRAAF and Herman R.J. VOLLEBERGH: <u>Burn or Bury? A Social Cost Comparison of Final Waste</u> Disposal Methods
ETA	47.2003	Jens HORBACH: Employment and Innovations in the Environmental Sector: Determinants and Econometrical Results for Germany
CLIM	48.2003	Lori SNYDER, Nolan MILLER and Robert STAVINS: The Effects of Environmental Regulation on Technology Diffusion: The Case of Chlorine Manufacturing
CLIM	49.2003	Lori SNYDER, Robert STAVINS and Alexander F. WAGNER: Private Options to Use Public Goods. Exploiting
CTN	50.2003	László Á. KÓCZY and Luc LAUWERS (lxi): The Minimal Dominant Set is a Non-Empty Core-Extension
CTN	51 0000	
CTN	51.2003	Matthew O. JACKSON (1x1): Allocation Rules for Network Games
CTN	52.2003	Ana MAULEON and Vincent VANNETELBOSCH (Ixi): <u>Farsightedness and Cautiousness in Coalition Formation</u>
CTN	53.2003	Fernando VEGA-REDONDO (lxi): Building Up Social Capital in a Changing World: a network approach
CTN	54.2003	Matthew HAAG and Roger LAGUNOFF (lxi): On the Size and Structure of Group Cooperation
CTN	55.2003	<i>Taiji FURUSAWA and Hideo KONISHI</i> (lxi): <u>Free Trade Networks</u>
CTN	56.2003	Halis Murat YILDIZ (lxi): National Versus International Mergers and Trade Liberalization
CTN	57.2003	Santiago RUBIO and Alistair ULPH (lxi): <u>An Infinite-Horizon Model of Dynamic Membership of International</u> Environmental Agreements
KNOW	58.2003	Carole MAIGNAN, Dino PINELLI and Gianmarco I.P. OTTAVIANO: <u>ICT, Clusters and Regional Cohesion: A</u> Summary of Theoretical and Empirical Research
KNOW	59 2003	Giorgio BELLETTINI and Gionmarco LP OTTAVIANO: Special Interests and Technological Change
FTA	60 2003	Romie SCHÖB: The Double Dividend Hynothesis of Environmental Taxes: A Survey
CLIM	61 2003	Michael FINUS, Fiko van FERLAND and Robert DELLINK, Stability of Climate Coalitions in a Cartel
CEIM	01.2005	Formation Game
GG	62 2003	Michael FINI'S and Rianca RUNDSHAGEN: How the Rules of Coalition Formation Affect Stability of
00	02.2005	International Environmental Agraements
SIEV	63 2003	Alberto DETRUCCI: Taxing Land Bent in an Open Economy
	64 2003	Aberto F E ALOV, Sector BADDETT, and Debert N. STAUDIS, Thirteen Due Ones A Compension of
CLINI	04.2003	Joseph E. ALDT, Scott BARRETT and Robert N. STAVINS. Initeen Plus One: A Comparison of
ares /	(5.000	Global Climate Policy Architectures
SIEV	65.2003	Edi DEFRANCESCO: The Beginning of Organic Fish Farming in Italy
SIEV	66.2003	Klaus CONRAD: Price Competition and Product Differentiation when Consumers Care for the
		Environment
SIEV	67.2003	Paulo A.L.D. NUNES, Luca ROSSETTO, Arianne DE BLAEIJ: Monetary Value Assessment of Clam
		Fishing Management Practices in the Venice Lagoon Results from a Stated Choice Exercise
CLIM	68.2003	ZhongXiang ZHANG: Open Trade with the U.S. Without Compromising Canada's Ability to Comply
mom	(0.000-	with its Kyoto 1 arget
KNOW	69.2003	David FRANTZ (IIX): Lorenzo Market between Diversity and Mutation
KNOW	70.2003	Ercole SORI (lix): Mapping Diversity in Social History
KNOW	71.2003	<i>Ljiljana DERU SIMIC</i> (lxii): What is Specific about Art/Cultural Projects?
KNOW	72.2003	Natalva V. TARANOVA (1xii): The Role of the City in Fostering Intergroup Communication in a
		Multicultural Environment: Saint-Petersburg's Case
KNOW	73 2002	Kristing CRANE (lyij). The City as an Arana for the Expression of Multiple Identities in the Ara of
	13.2003	Chile Liestien and Mientian
WNOW	74 2002	Giodalisation and Migration
KNOW	/4.2003	Kazuma MATOBA (Ix11): Glocal Dialogue- Transformation through Transcultural Communication
KNOW	75.2003	Catarina REIS OLIVEIRA (lxii): Immigrants' Entrepreneurial Opportunities: The Case of the Chinese
		in Portugal
KNOW	76.2003	Sandra WALLMAN (lxii): The Diversity of Diversity - towards a typology of urban systems

77.2003 Richard PEARCE (lxii): <u>A Biologist's View of Individual Cultural Identity for the Study of Cities</u>

78.2003 Vincent MERK (lxii): Communication Across Cultures: from Cultural Awareness to Reconciliation of the Dilemmas

KNOW

79.2003 *Giorgio BELLETTINI, Carlotta BERTI CERONI and Gianmarco I.P.OTTAVIANO*: Child Labor and Resistance to Change

(1) This paper was presented at the Workshop "Growth, Environmental Policies and Sustainability" organised by the Fondazione Eni Enrico Mattei, Venice, June 1, 2001

(li) This paper was presented at the Fourth Toulouse Conference on Environment and Resource Economics on "Property Rights, Institutions and Management of Environmental and Natural Resources", organised by Fondazione Eni Enrico Mattei, IDEI and INRA and sponsored by MATE, Toulouse, May 3-4, 2001

(lii) This paper was presented at the International Conference on "Economic Valuation of Environmental Goods", organised by Fondazione Eni Enrico Mattei in cooperation with CORILA, Venice, May 11, 2001

(liii) This paper was circulated at the International Conference on "Climate Policy – Do We Need a New Approach?", jointly organised by Fondazione Eni Enrico Mattei, Stanford University and Venice International University, Isola di San Servolo, Venice, September 6-8, 2001

(liv) This paper was presented at the Seventh Meeting of the Coalition Theory Network organised by the Fondazione Eni Enrico Mattei and the CORE, Université Catholique de Louvain, Venice, Italy, January 11-12, 2002

(lv) This paper was presented at the First Workshop of the Concerted Action on Tradable Emission Permits (CATEP) organised by the Fondazione Eni Enrico Mattei, Venice, Italy, December 3-4, 2001 (lvi) This paper was presented at the ESF EURESCO Conference on Environmental Policy in a Global Economy "The International Dimension of Environmental Policy", organised with the collaboration of the Fondazione Eni Enrico Mattei, Acquafredda di Maratea, October 6-11, 2001

(lvii) This paper was presented at the First Workshop of "CFEWE – Carbon Flows between Eastern and Western Europe", organised by the Fondazione Eni Enrico Mattei and Zentrum fur Europaische Integrationsforschung (ZEI), Milan, July 5-6, 2001

(lviii) This paper was presented at the Workshop on "Game Practice and the Environment", jointly organised by Università del Piemonte Orientale and Fondazione Eni Enrico Mattei, Alessandria, April 12-13, 2002

(lix) This paper was presented at the ENGIME Workshop on "Mapping Diversity", Leuven, May 16-17, 2002

(lx) This paper was presented at the EuroConference on "Auctions and Market Design: Theory, Evidence and Applications", organised by the Fondazione Eni Enrico Mattei, Milan, September 26-28, 2002

(lxi) This paper was presented at the Eighth Meeting of the Coalition Theory Network organised by the GREQAM, Aix-en-Provence, France, January 24-25, 2003

(lxii) This paper was presented at the ENGIME Workshop on "Communication across Cultures in Multicultural Cities", The Hague, November 7-8, 2002

KNOW KNOW

2002 SERIES

CLIM	Climate Change Modelling and Policy (Editor: Marzio Galeotti)
VOL	Voluntary and International Agreements (Editor: Carlo Carraro)
SUST	Sustainability Indicators and Environmental Valuation (Editor: Carlo Carraro)
NRM	Natural Resources Management (Editor: Carlo Giupponi)
KNOW	Knowledge, Technology, Human Capital (Editor: Dino Pinelli)
MGMT	Corporate Sustainable Management (Editor: Andrea Marsanich)
PRIV	Privatisation, Regulation, Antitrust (Editor: Bernardo Bortolotti)
ETA	Economic Theory and Applications (Editor: Carlo Carraro)

2003 SERIES

CLIM	Climate Change Modelling and Policy (Editor: Marzio Galeotti)
GG	Global Governance (Editor: Carlo Carraro)
SIEV	Sustainability Indicators and Environmental Valuation (Editor: Anna Alberini)
NRM	Natural Resources Management (Editor: Carlo Giupponi)
KNOW	Knowledge, Technology, Human Capital (Editor: Gianmarco Ottaviano)
IEM	International Energy Markets (Editor: Anil Markandya)
CSRM	Corporate Social Responsibility and Management (Editor: Sabina Ratti)
PRIV	Privatisation, Regulation, Antitrust (Editor: Bernardo Bortolotti)
ETA	Economic Theory and Applications (Editor: Carlo Carraro)
CTN	Coalition Theory Network